CEBAF EXPERIMENT 91-003A Study of Longitudinal Charged Pion Electroproductionin D2, He3, and He4Harold E. Jackson (Spokesperson)Physics Bldg 203, Argonne National Laboratory9700 S. Cass Ave., Argonne, IL 60439E-mail: BO4353@anlvmPh: (708) 252-4013Fax: (708) 252-3903K. Coulter, D.F. Geesaman, R.J. Holt, S. Kaufman,D. Potterveld, B. ZeidmanArgonne National LaboratoryR. GilmanRutgers UniversityB. KinneyUniversity of ColoradoJ. Mougey, B. SaghaiCentre d�Etudes Nucleaire de SaclayR. RegelNorthwestern UniversityExperiment approved for 504 hours in Hall C